Author's response to reviews

Title: Antibiotic prophylaxis for caesarean section at a Ugandan Hospital: a randomised clinical trial evaluating the effect of administration time on the incidence of post-operative infections

Authors:

Lomangisi D Dlamini (ngisi2@yahoo.com)
Musa Sekikubo (msekikubo@yahoo.com)
Janat Tumukunde (janattumukunde@gmail.com)
Charles Kojjo (charleskojjo@gmail.com)
Davidson Ocen (d4davidson@gmail.com)
Agnes Wabule (aggiewabule2000@yahoo.co.uk)
Arthur Kwizera (akwizera@chs.mak.ac.ug)

Version: 7 Date: 11 March 2015

Author's response to reviews:

The Editorial Team
Thank for your continuous review of our manuscript.
We have done the recommended revisions on the paper.
Trial registration details have been included on the online abstract.
The registry website is www.pactr.org and we have also realised it is a challenge to search the register with the registration number it is easier with the title. We apologies that there is no direct link to access the study on the register. We have double checked the number is correct.

Yours sincerely

Dr Lomangisi Diana Dlamini